Author's response to reviews

Title: Unbearable suffering in end-of-life cancer patients A cross-sectional study in primary care in the Netherlands

Authors:

Cees DM Ruijs (c.ruys3@kpnplanet.nl)
Ad JFM Kerkhof (ajfm.kerkhof@vu.nl)
Gerrit van der Wal (g.vanderwal@vumc.nl)
Bregje D Onwuteaka-Philipsen (b.philipsen@vumc.nl)

Version: 2 Date: 27 February 2012

Author's response to reviews:

To:
Hayley Henderson, Biomed Central

Dear editor,

conform the advise of Ursula D'Souza, senior editor of BMC Medicine, we have adapted the paper in general, conform the STROBE Statement-Checklist of items that should be included in reports of cross-sectional studies. Also the title has been changed.